

Telacebec (Q203)-containing intermittent oral regimens sterilized mice infected with Mycobacterium ulcerans after only 16 doses

Aurélie Chauffour, Jérôme Robert, Nicolas Veziris, Alexandra Aubry, Kevin Pethe, Vincent Jarlier

▶ To cite this version:

Aurélie Chauffour, Jérôme Robert, Nicolas Veziris, Alexandra Aubry, Kevin Pethe, et al.. Telacebec (Q203)-containing intermittent oral regimens sterilized mice infected with Mycobacterium ulcerans after only 16 doses. PLoS Neglected Tropical Diseases, 2020, 14 (8), pp.e0007857. 10.1371/journal.pntd.0007857. hal-02949442

HAL Id: hal-02949442 https://hal.sorbonne-universite.fr/hal-02949442

Submitted on 25 Sep 2020

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Citation: Chauffour A, Robert J, Veziris N, Aubry A, Pethe K, Jarlier V (2020) Telacebec (Q203)containing intermittent oral regimens sterilized mice infected with *Mycobacterium ulcerans* after only 16 doses. PLoS Negl Trop Dis 14(8): e0007857. https://doi.org/10.1371/journal. pntd.0007857

Editor: Abdallah M. Samy, Faculty of Science, Ain Shams University (ASU), EGYPT

Received: October 17, 2019

Accepted: June 15, 2020

Published: August 31, 2020

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pntd.0007857

Copyright: © 2020 Chauffour et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: All relevant data are within the manuscript and its Supporting Information files.

RESEARCH ARTICLE

Telacebec (Q203)-containing intermittent oral regimens sterilized mice infected with *Mycobacterium ulcerans* after only 16 doses

Aurélie Chauffour^{1*}, Jérôme Robert^{1,2}, Nicolas Veziris^{1,2,3}, Alexandra Aubry^{1,2}, Kevin Pethe^{4,5}, Vincent Jarlier^{1,2}

1 Sorbonne Université, INSERM, U1135, Centre d'Immunologie et des Maladies Infectieuses (CIMI-Paris), Paris, France, 2 Centre National de Référence des Mycobactéries et de la Résistance des Mycobactéries aux Antituberculeux, Laboratoire de Bactériologie-Hygiène, Groupe hospitalier APHP, Sorbonne Université, Site Pitié-Salpêtrière, Paris, France, 3 Département de Bactériologie, Groupe hospitalier APHP, Sorbonne Université, Site Saint-Antoine, Paris, France, 4 Lee Kong Chian School of Medicine, Nanyang Technological University, Experimental Medicine Building, Singapore, Singapore, 5 School of Biological Sciences, Nanyang Technological University, Singapore, Singapore

* aurelie.chauffour@sorbonne-universite.fr

Abstract

Buruli ulcer (BU), caused by *Mycobacterium ulcerans*, is currently treated with a daily combination of rifampin and either injectable streptomycin or oral clarithromycin. An intermittent oral regimen would facilitate treatment supervision. We first evaluated the bactericidal activity of newer antimicrobials against *M. ulcerans* using a BU animal model. The imidazopyridine amine telacebec (Q203) exhibited high bactericidal activity whereas tedizolid (an oxazolidinone closely related to linezolid), selamectin and ivermectin (two avermectine compounds) and the benzothiazinone PBTZ169 were not active. Consequently, telacebec was evaluated for its bactericidal and sterilizing activities in combined intermittent regimens. Telacebec given twice a week in combination with a long-half-life compound, either rifapentine or bedaquiline, sterilized mouse footpads in 8 weeks, i.e. after a total of only 16 doses, and prevented relapse during a period of 20 weeks after the end of treatment. These results are very promising for future intermittent oral regimens which would greatly simplify BU treatment in the field.

Author summary

The current treatment for Buruli ulcer (BU), an infection caused by *Mycobacterium ulcerans*, is based on a daily antibiotic combination of rifampin associated with streptomycin or clarithromycin. A shorter or intermittent treatment without an injectable drug would clearly simplify the management in the field. We evaluated the bactericidal activity of several new antimicrobial drugs in a mouse model of BU and found that telacebec (Q203) exhibited the greatest bactericidal effect. We subsequently identified new antibiotic combinations containing telacebec with high sterilizing activity when administered twice a week for 8 weeks, i.e. at a total of only 16 doses.

Funding: This work was funded by the Raoul Follereau Foundation. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: The authors have declared that no competing interests exist.

Introduction

Buruli ulcer (BU), caused by *Mycobacterium ulcerans*, was only treated by surgery until 2004 [1]. The first medical treatment recommended by the World Health Organization (WHO) was an eight-week daily treatment based on an association of two antibiotics, rifampin (RIF), an oral ansamycin, and streptomycin (STR), an injectable aminoglycoside [1]. Currently a promising fully oral regimen combining RIF and clarithromycin (CLR), a macrolide compound [2,3], is tested clinically on a large scale (NCT01659437, clinicaltrials.gov).

The oral RIF-CLR combination was promoted to eliminate the toxic effects and injections of aminoglycosides, resulting in greater patient adherence and safety. Nevertheless, this combination is given daily for eight weeks. Shorter or intermittent treatment would facilitate adherence as well as supervision by healthcare workers. For instance, many Buruli ulcer patients with small-to-moderate size wounds are on ambulatory care and visit healthcare centres twice or three times per week for dressing changes, a rhythm that could allow receiving supervised intermittent antibiotic administration [4].

Our main objective was to identify alternative oral regimens active against BU by using a validated BU animal model. As a first step, we screened several new drug candidates for their *in vivo* bactericidal activity against *M. ulcerans*. Based on available data on their activity against *M. tuberculosis* or *M. ulcerans*, the following compounds, with either short or long half-life, were selected as potentially interesting: selamectin (SEL) and ivermectin (IVE), two drugs from the avermectin family with antiparasitic properties [5–7]; tedizolid (TDZ) [8], a new oxazolidinone active against *M. tuberculosis* that has the same mechanism of action as linezolid (LZD), a drug active against *M. tuberculosis* and *M. ulcerans* [9] with high bioavailability and a long half-life [10]; the 2-piperazino-benzothiazinone 169 (PBTZ), shown to be highly active against *M. tuberculosis* [11,12]; the imidazopyridine amine telacebec (Q203) that targets the respiratory terminal oxidase cytochrome bc1:aa3 in *M. tuberculosis* and that was recently shown to prevent mortality and reduce CFU counts in the footpads of mice infected with *M. ulcerans* [13] and to exhibit sterilizing activity when administrated in combination with other drugs [14].

Because the first-step screening demonstrated that Q203 was the compound with the highest bactericidal activity, and because of the long half-life of this compound [15], we measured, in a second experiment, the bactericidal and sterilizing activity of intermittent regimens containing Q203 combined with rifapentine or bedaquiline, other antibiotics with a long half-life and known to be active against *M. ulcerans* [16,17].

Methods

Infection of mice with M. ulcerans

In the first and the second experiment, respectively, 190 and 390 4-week-old female BALB/c/j mice were used (Janvier Labs, Le Genest Saint-Isle, France). Mice were inoculated, according to Shepard [18] in the left hind footpad with 0.03 ml of a bacterial suspension containing 5 \log_{10} bacilli of the *M. ulcerans* strain Cu001. The number of live bacilli in the bacterial suspension was determined to be 5.02 and 4.6 \log_{10} in the first and second experiment, respectively, by culturing the inoculum on Lowenstein-Jensen (LJ) medium. This strain, isolated in 1996 from a BU patient in Adzopé, Ivory Coast [19], was kindly provided by the local laboratory, blinded to patient identity. The strain is susceptible to all drugs used in BU treatment and was maintained in our laboratory through regular mouse footpad passage.

Treatment of mice

Treatment was initiated when the infection was well established, *i.e.* at a footpad swelling index of "2" (inflammatory swelling limited to the inoculated footpad) to "3" (inflammatory swelling involving the entire inoculated footpad) on a 4-grade scale [19] (Fig 1). This stage of infection was reached six weeks after the inoculation. Mice were randomly allocated into eight groups (1st experiment) and ten groups (2nd experiment) using a randomization table generated by the web site Randomization.com (https://www.randomization.com).

The groups were treated (with respect to drug, dosage, as well as number of doses/number of weeks, respectively) as follows:

In the first experiment, we used 8 groups including one untreated control group of 30 mice and seven antibiotic-treated groups of 20 mice, each treated with either TDZ, 10 mg/kg, 5/7; LZD, 100 mg/kg, 5/7; SEL, 12 mg/kg, 1/7; IVE, 1 mg/kg, 5/7; Q203, 5 mg/kg, 5/7 or PBTZ, 25 mg/kg, 5/7 and, as treatment control, RIF, 10 mg/kg, 5/7 alone or combined with STR, 150 mg/kg.

In the second experiment, out of 10 groups in total, we used 6 groups including one untreated control group of 27 mice and five antibiotic-treated groups of 27 mice, each receiving monotherapy with RIF, 10 mg/kg, 5/7; rifapentine (RPT), 20 mg/kg, 2/7; bedaquiline (BDQ), 25 mg/kg, 2/7; or Q203, 5 mg/kg, 5/7 or 2/7 while we used 4 groups of 57 mice each treated with drug combinations (dosages as above), i.e. Q203-RIF, 5/7; Q203-RPT, 2/7; Q203-BDQ, 2/7 or RIF-CLR, 100 mg/kg, 5/7 as control.

LZD was purchased from Pfizer, France; SEL and IVE from Merck, France; RIF from Sandoz, France; STR from Panpharma, France; and CLR from Abbott, France. TDZ was kindly





Fig 1. Images representative of the lesion indexes used (A) index 0; (B) index 1; (C) index 2; (D) index 3; (E) index 4. Measure of the index was taken at the middle of the footpad (see the black bracket). Measures of indexes were as follows: 2mm for index 0; 3mm for index 1; 4.5mm for index 2; 5.5mm for index 3; and 6.5mm for index 4.

https://doi.org/10.1371/journal.pntd.0007857.g001

provided by MSD-MERCK group, BDQ by Janssen Pharmaceutica, Belgium, and PBTZ by Stewart Cole (Global Health Institute, Ecole Polytechnique Fédérale de Lausanne, Lausanne, Switzerland; Institut Pasteur, rue du Docteur Roux, France). Q203 was custom-synthetized at GVKBio (GVK Biosciences Private Limited, Hyderabad, India) and contributed by Kevin Pethe.

Antibiotics were suspended in 0.05% agar-containing distilled water, except for STR, which was dissolved in buffered saline, Q203 in 1% DMSO-20% TPGS (D- α -Tocopherol polyethylene glycol 1000 succinate) and PBTZ in a solution of 1% carboxyl-methylcellulose-1% Tween 80. BDQ was provided by Janssen Pharmaceutica in a 20% hydropropyl- β -cyclodextrin formulation.

Treatments were administered for 4 or 8 weeks in both experiments and all drugs were orally administered by gavage in a final volume of 0.2 ml, except STR, which, in the same volume, was injected subcutaneously.

Assessment of M. ulcerans infection and effectiveness of treatment

Two methods were used for assessing the development of *M. ulcerans* infection and the effectiveness of treatments: (i) a clinical method consisting in weekly evaluation of the lesion index as previously described [19] and (ii) a bacteriological method consisting in the cultivation of M. ulcerans from the mouse footpads. When footpad swelling totally regresses under treatment, some redness often remains, indicative of scar tissue. We chose to assign index "1" to this stage, as index "0" is assigned to a normal footpad. For CFU enumeration in footpads, mice were sacrificed by cervical dislocation as recommended by the European directive 2010/63 and the French decree n°2013-118. Mouse footpads were removed aseptically and ground in 2 ml of Hank's balanced salt solution in a tissue grinder (Octo Dissociator GentleMACS, Miltenyi Biotec, France). Suspensions were then plated onto LJ slants containing vancomycin (10 µg/ml), colistin (40 µg/ml) and amphotericin B (10 µg/ml) to limit contamination. In the first experiment, for the untreated group and the groups treated with RIF, TDZ, LZD, SEL, IVE or PBTZ, suspensions were serially 10-fold diluted to 10^{-4} , and 0.1 ml of the dilutions was plated in duplicate onto LJ media. For the Q203 and the RIF-STR-treated groups, the entire volume of the footpad suspensions was plated onto LJ-media, at 0.2 ml per slant. In the second experiment, suspensions were serially diluted 10-fold to 10^{-4} for the untreated group, to 10^{-3} for the RIF and RPT groups at 2, 4 and 8 weeks and for the BDQ group at 2 weeks; to 10^{-2} for the Q203 2/7, Q203 5/7 and RIF-CLR groups at 2 weeks, and for the BDQ group at 4 weeks. A volume of 0.1 ml of each of these serial dilutions was plated in duplicate onto LJ media. For all the remaining time points, including those for all Q203 combinations, the entire volume of the footpad suspensions was plated onto LJ media, at 0.2 ml per slant. All tubes were incubated at 30°C for 90 days.

In the first experiment, the footpad lesion indexes were determined weekly during the 8-week period of treatment and CFUs were counted at weeks 4 and 8. In the 2^{nd} experiment, footpad lesion indexes were determined weekly during the 8-week period of treatment and CFUs were counted at weeks 2, 4 and 8. Moreover, in the latter experiment, 30 mice that had been treated for 8 weeks with combination therapies were held without treatment during an additional period of 20 weeks to monitor relapses of *M. ulcerans* infection. Lesion indexes were determined weekly during this period and CFUs were counted at the end of the observation period.

MIC determination

In order to assess a possible acquisition of resistance to the antibiotics used during treatment, MICs of the antibiotics used for the treatment were determined for the bacilli isolated from

relapsing mice during the observation period and for the initial strain Cu001. The isolates were suspended in distilled water to match the turbidity of a Mc Farland 3 standard. RIF and CLR were tested on Middlebrook 7H11 + 10% OADC (Oleicacid-Albumin-Dextrose-Catalase) medium (pH 7.4) and CLR was also tested on Mueller Hinton medium (pH 6.6). RIF was dissolved in dimethylformamide and CLR in distilled water, and both were twofold diluted in their respective solvent and incorporated into the culture media to obtain final concentrations ranging from 4 to 0.12 µg/ml. Suspensions (0.1 ml) of two distinct bacillus isolates (pure and diluted 10^{-2}) were plated onto drug-containing and drug-free media used for growth control. All media were incubated at 30°C and examined after 60 and 90 days. The minimum inhibitory concentration (MIC) was defined as the lowest concentration of the drug preventing \geq 99% of the growth on drug-free medium [20].

Statistical analysis

The Mann–Whitney *U* test was used to analyze the results (BiostaTGV). Relapse rates were also compared using Fisher's exact test. A p-value < 0.05 was considered to be statistically significant. A significant decrease in the mean CFU value per footpad of treated as compared to untreated mice was used to assess the bactericidal effect of a given drug regimen.

Ethics statement

The experimental project was favorably evaluated by the ethics committee n°005 Charles Darwin localized at the Pitié-Salpêtrière Hospital and clearance was given by the French Ministry of Education and Research under the number APAFIS#9576–20170301171176185 v2. Our animal facility received the authorization to carry out animal experiments (license number C-75-13-01) on April 27, 2017. The persons who carried out the animal experiments had followed a specific training recognized by the French Ministry of Education and Research.

Results

First experiment

The mean lesion index (MLI) was 2.8±0.66 at the start of treatment (Fig 2). Untreated control mice with swollen footpads and MLIs that increased from 2.8 to 4 after 2 weeks, and mice with extensive lesions (inflammatory swelling of the whole limb, accompanied by a deteriorating general condition) either died or had to be sacrificed at week 4. MLIs in RIF-treated control groups increased to 3.8 after one week and then decreased to remain stable at 3.4, while decreasing to 2 in the RIF-STR group. MLIs in the SEL-, IVE-, TDZ- and PBTZ-treated groups continued to increase and mice either died or had to be sacrificed at week 4 due to advanced lesions comparable to those observed in untreated mice. In contrast, the MLI in the Q203-treated group decreased to 1.9 after 1 week of treatment and to 1.2 after 8 weeks.

All untreated mice had culture-positive footpads at the beginning of treatment, with a mean of $6.93 \pm 0.20 \log_{10}$ CFUs, a count that remained stable at week 4 (Table 1). All mice remained culture-positive after 4 weeks of treatment in the RIF control groups but the mean CFU counts were significantly lower (p < 0.05) than those in the untreated mouse group ($4.58 \pm 1.29 \log_{10}$ for the RIF-(p = 0.0007) and $2.15 \pm 1.20 \log_{10}$ for the RIF-STR (p = 0.00006) treated groups, respectively). After 8 weeks of treatment, only 5 out of 8 mice in the RIF groups remained culture-positive with a low mean CFU count ($<1 \log_{10}$). After 4 weeks of treatment, all mice were culture-positive in the TDZ, SEL, IVE and PBTZ groups, with CFU counts not statistically different from those in the untreated control group. Due to advanced footpad lesion and mortality of mice, the evaluation point initially planned at week 8 was cancelled for



Fig 2. First experiment: Mean lesion index of *M. ulcerans*-infected mouse footpads during treatment for 8 weeks. Doses were as follow: rifampin (RIF), 10 mg/kg; streptomycin (STR), 150 mg/kg; tedizolid (TDZ), 10 mg/kg; linezolid (LZD), 100 mg/kg; selamectin (SEL), 12 mg/kg; ivermectin (IVE), 1 mg/kg; telacebec (Q203), 5 mg/kg; PBTZ169 (PBTZ), 25 mg/kg. The number of doses/number of weeks for RIF alone were 5/7; for RIF-STR, 5/7; for TDZ, 5/7; for LZD, 5/7; for SEL, 1/7; for IVE, 5/7; for Q203, 5/7 and for PBTZ, 5/7.

https://doi.org/10.1371/journal.pntd.0007857.g002

these groups. After 4 weeks of treatment, 3 out of 10 mice were culture-negative in the Q203-treated group with a low mean CFU count $(1.14 \pm 1.30 \log_{10})$, i.e. different from those in the RIF-treated group (p = 0.00002). After 8 weeks of treatment, all mice were culture-negative in the Q203 group.

Second experiment

The MLI was 3±0.79 at the start of the treatment (**Fig 3**). The footpad MLIs of untreated control mice increased from 3 to 4 after 2 weeks and the mice had to be sacrificed at week 4 due to extensive BU lesions. The MLIs in the RIF-, RPT- and BDQ-treated groups increased to 3.8–4 after one week of treatment and then decreased, by week 8, to 3.6 (BDQ), 3 (RIF) and 2.7 (RPT). In the Q203-treated groups (*n* of doses/*n* of weeks, 5/7 or 2/7), MLIs increased slightly to 3.5 after one week of treatment and then decreased rapidly to 1.6–1.7 at week 8. The MLI in the group treated with RIF-CLR increased to 4 after 1 week of treatment, then decreased to 1.4 at week 12 and remained stable until week 20, but increased again to 2.6 at week 28. The MLI in the group treated with Q203-BDQ increased slightly to 3.6 after one week of treatment and decreased thereafter to 1.4 at week 12, a level that remained stable until week 28. The MLIs in the groups treated with Q203 combined with RIF or RPT decreased rapidly to 1.2–1.3 at week 8 and remained at that level until week 28.

All untreated mice had culture-positive footpads at the beginning of treatment, with a mean of $6.87 \pm 0.10 \log_{10}$ CFUs, a count that remained stable at weeks 2 and 4 (Table 2). All treated mice remained culture-positive after 2 weeks of treatment but the CFU counts were significantly lower in all treated groups compared to those in the untreated group. Moreover, CFU counts were lower in the RPT (compared to RIF *p* = 0.008 and *p* = 0.0003 compared to BDQ), Q203 (for Q203 2/7 compared to RIF *p* = 0.008; compared to BDQ *p* = 0.001; for Q203

Regimen ^a (<i>n</i> doses/ <i>n</i> weeks)	Results during treatment									
		Day 0		Week 4	Week 8					
	Culture positivity rate	Mean (±SD) CFU per group	Culture positivity rate	Mean (±SD) CFU per group	Culture positivity rate	Mean (±SD) CFU per group				
Untreated control	10/10	6.93±0.20	13/13 ^b	6.81±0.36						
RIF 5/7			9/9 ^c	4.58±1.29	5/8 ^c	0.47±1.16				
RIF-STR 5/7			8/8 ^d	2.15±1.20	3/9 ^d	0.57±0.90				
TDZ 5/7			7/7 ^e	6.35±0.59	f					
LZD 5/7			10/10	2.83±0.62	9/10	1.87±1.11				
SEL 1/7			4/4 ^g	6.24±0.82	f					
IVE 5/7			8/8 ^h	6.30±0.82	f					
Q203 5/7			7/10	1.14±1.30	0/10					
PBTZ 5/7			7/7 ⁱ	7.08±0.64	f					

Table 1. First experiment: Results of footpad cultures during the treatment of mice infected with M. ulcerans.

^a: treatment began 6 weeks after inoculation with 5.02 log₁₀ bacilli per footpad when the infected swollen footpads had reached a lesion index of 2.8. Drugs were administered 5-times a week, except for selamectin (SEL) which was administered once a week. Dosages were as follows: rifampin (RIF), 10 mg/kg; streptomycin (STR), 150 mg/kg; tedizolid (TDZ), 10 mg/kg; linezolid (LZD), 100 mg/kg; SEL, 12 mg/kg; ivermectin (IVE), 1 mg/kg; telacebec (Q203), 5 mg/kg; PBTZ169 (PBTZ), 25 mg/kg.

^b: due to advanced lesions, all mice from the untreated control group were in fact sacrificed at week 3; nevertheless, footpad cultures were contaminated in 7 out of 20 mice.

^c: 3 mice died in the RIF groups due to an accident during gavage.

^d: 3 mice died in the RIF-STR groups due to an accident during gavage.

^e: footpad cultures were contaminated due to advanced lesions in 3 mice.

^f: cultures at week 8 were not performed because of advanced necrotized footpad lesions.

^g: 5 mice died from *M.ulcerans* infection and footpad cultures were contaminated due to advanced lesion in 1 mouse.

^h: 1 mouse died from *M.ulcerans* infection and footpad cultures were contaminated due to advanced lesion in 1 mouse.

ⁱ: footpad cultures were contaminated due to advanced lesions in 3 mice.

https://doi.org/10.1371/journal.pntd.0007857.t001

5/7 compared to RIF p = 0.01 and p = 0.0001 compared to BDQ) and the 4 combined-treatment groups than in the RIF and BDQ groups (p-values for combined treatment groups vs RIF: CR p = 0.003; QR p = 0.002; QB p = 0.006; QP p = 0.002; all p-values for combined treatment groups vs BDQ were 0.0002). After 4 weeks of treatment, some of the mice became culture-negative, especially in the groups treated with Q203-RPT and Q203-BDQ in which the CFU counts were very low (~0.2 log₁₀). CFU counts at 4 weeks remained lower in the RPT and Q203 than in the RIF (p = 0.006, p = 0.04 and p = 0.002 for RPT, Q203 2/7 and Q203 5/7 respectively) and BDQ groups (p = 0.007, p = 0.001 and p = 0.001 for RPT, Q203 2/7 and Q203 5/7 respectively) and they were lower in the Q203-RPT (p = 0.04 and p = 0.03) and Q203-BDQ (p = 0.03 and p = 0.01) groups than in the Q203-RIF and RIF-CLR groups. After 8 weeks of treatment, all mice treated with Q203 alone or combined with RIF (5/7) or RPT (2/7) or BDQ (2/7) became culture-negative. Few mice remained culture-positive in the RIF, RPT and RIF-CLR groups with very low mean CFU counts (<0.5 log₁₀) but all remained culture-positive in the BDQ group (1.28 ± 0.93 log₁₀).

During the 20-week observation period after stopping the treatment, no clinical or bacteriological relapse was observed in the 3 groups treated with Q203 combinations whereas 8 out of 26 mice treated with RIF-CLR relapsed with a mean CFU count of $0.87 \pm 0.93 \log_{10}$. This proportion was significantly higher than those in other groups (p = 0.001).



Fig 3. Second experiment: Mean lesion index of M. ulcerans-infected mouse footpads during and after treatment for 8 weeks. Dosages were as follows: rifampin (RIF), 10 mg/kg; rifapentine (RPT), 20 mg/kg; bedaquiline (BDQ), 25 mg/kg; Q203, 5 mg/kg; clarithromycin (CLR), 100 mg/kg. RIF was given (n of doses/n of weeks) 5/7; RPT, 2/7; BDQ, 2/7; Q203, 2/7 or 5/7; RIF-CLR, 5/7; Q203-RIF, 5/7; Q203-BDQ, 2/7 and Q203-RPT, 2/7. 1.2–1.3 at week 8 and remained at that level until week 28.

https://doi.org/10.1371/journal.pntd.0007857.g003

MICs for M. ulcerans bacilli recovered from relapsing mice

The MICs for bacilli isolated from the 8 relapsing mice in the RIF-CLR-treated group remained stable when compared to the initial MICs of RIF ($0.5-1 \mu g/ml$) and CLR ($0.5 \mu g/ml$) found for *M. ulcerans* Cu001 (the MICs of the latter were found to be the same on the 7H11 and MH media).

Discussion

Although nearly all Buruli ulcer lesions can be successfully treated by a two-month antibiotic combination regimen administered daily [21,22], new shorter and/or intermittent regimens would greatly simplify the treatment procedure in the field. Indeed, intermittent treatment would allow for organizing direct supervision of therapy and for combining drug administration and change of dressings in remote places where BU is prevalent.

The key finding of the present work was that regimens combining telacebec with rifapentine or bedaquiline, two long lasting drugs, sterilized infected mouse footpads after twice weekly administration during for 8 weeks, i.e. after a total of only 16 doses.

Our first *in vivo* screening experiment aimed at identifying newer bactericidal drugs. Indeed, available data on the activity of several new drugs against *M. tuberculosis* or *M. ulcerans* justified a systematic evaluation in a BU mouse model that has been successfully used for many years for this purpose [9]. IVE, SEL and TDZ were not bactericidal after 4 weeks of treatment and failed to prevent mortality during 8 weeks. The doses used in our experiment were taken from available pharmacokinetic data. IVE, a long-lasting drug in humans (half-life, 15-19h) and in mice (9h), when used in mice at a dose of 0.2 mg/kg, was shown to yield serum

Regimen ^a (n doses/n weeks)	Results during treatment									Results after observation period	
	Day 0		Week 2		Week 4		Week 8		Week 28		
	Culture positivity rate	Mean (±SD) CFU per group									
Untreated control	9/9	6.87±0.10	8/8 ^b	7.00±0.16	6/6 ^b	6.84±0.35					
RIF 5/7			9/9	5.71±0.65	6/6 ^c	2.78±0.81	2/8 ^c	0.44±0.81			
RPT 2/7			8/9	4.18±1.71	3/9	0.63±1.03	1/9	0.20±0.59			
BDQ 2/7			7/7 ^d	6.12±0.31	7/7 ^d	5.48±0.14	5/5 ^d	1.28±0.93			
Q203 2/7			9/9	4.88±0.42	5/9	1.42±1.21	0/9				
Q203 5/7			9/9	4.90±0.29	7/9	0.74±0.79	0/9				
RIF-CLR 5/7			9/9	4.42±0.55	6/7 ^e	1.15±0.84	1/8 ^e	0.22±0.63	8/26 ^e	0.87±1.36	
Q203-RIF 5/ 7			9/9	4.52±0.36	9/9	1.20±1.04	0/8 ^f		0/30		
Q203-BDQ 2/7			9/9	4.65±0.75	2/9	0.17±0.51	0/9		0/30		
Q203-RPT 2/ 7			9/9	4.40±0.78	2/9	0.25±0.49	0/9		0/30		

Table 2. Second experiment: Results of footpad cultures during the treatment of mice infected with *M. ulcerans* (for 2, 4 or 8 weeks) and relapse rate after treatment completion.

^a: treatments were begun 6 weeks after inoculation of 4.6 log₁₀ CFUs per footpad when the infected swelling footpads reached a lesion index of 3. Drugs were administered two or five times a week and dosages were as follows: rifampin (RIF), 10 mg/kg; rifapentine (RPT), 20 mg/kg; bedaquiline (BDQ), 25mg/kg; telacebec

(Q203), 5 mg/kg and clarithromycin (CLR), 100 mg/kg.

^b: footpad cultures were contaminated due to advanced lesion in 1 mouse in the untreated Week 2 control group, and in 1 mouse in the Week 4 group.

c: footpad cultures were contaminated due to advanced lesions in 3 mice in the Week 4 RIF group, and in 1 mouse in the Week 8 group.

^d: footpad cultures were contaminated due to advanced lesions in 1 mouse in the Week 2 BDQ group, in 2 mice in the Week 4 group and in 2 mice in the Week 8 group; 2 mice died due to an accident during gavage in the Week 8 group.

^e: footpad cultures were contaminated due to advanced lesion in 2 mice in the Week 4 RIF-CLR group and in 3 mice in the relapse observation group; 1 mouse each died due to an accident during gavage in the Week 8 group and in the relapse observation group.

^f: 1 mouse died due to an accident during gavage in the Week 8 Q203-RIF group.

https://doi.org/10.1371/journal.pntd.0007857.t002

concentrations lower than those found in humans at standard therapeutic dosage [7]. We therefore used a higher dose (1 mg/kg), which also failed to control infection in our model. A similarly unfavorable result was obtained with SEL used at 12 mg/kg as proposed in a previous publication [23]. However, it has been suggested that these two avermectin compounds might be used safely at even higher doses [7], which could be evaluated in future studies. The dose of TDZ used in the present study, i.e. 10 mg/kg, was shown to yield pharmacokinetic profiles close to those observed in humans at the therapeutic dose of 200 mg [24–26]. Contrasting with the disappointing results obtained with TDZ, and as reported in a previous study [9], a marked bactericidal activity was obtained with LZD, an oxazolidinone included in the experiment as a positive control for TDZ. Surprisingly, PBTZ was not bactericidal in our BU model at 25 mg/kg, a dose shown to be active against *M. tuberculosis* in mice [11]. Yet in *M. ulcerans*, as well as in *M. tuberculosis*, DprE1, the target of benzothiazinones, carries a cysteine at position 387 in lieu of a serine or an alanine, which have been shown to confer natural resistance to PBTZ in *M. avium* or *M. aurum*, respectively [27]. Thus, the reason for the disappointing result obtained with PBTZ in our BU model is unclear.

RIF and RIF-STR were highly bactericidal as observed in all our preceding studies [2,28]. Q203 drastically reduced the lesion index and CFU counts after 4 weeks of treatment and all

mice became culture-negative after 8 weeks. These results obtained with Q203 were clearly superior to those obtained with the usual positive control with RIF alone, and even with the reference combination regimen RIF-STR.

Drugs with low bactericidal activity, when given as monotherapy, might be of interest when used in combination with other drugs. However, since our goal was to obtain the most effective combination regimen, we selected, for the second experiment, combinations of drugs shown to be highly active separately. The results of this experiment demonstrated that regimens combining Q203 with RIF or RPT or BDQ were not only bactericidal, making all footpads culture-negative after 8 weeks of treatment, but also sterilized them and prevented relapse during an observation period of 20 weeks after stopping the treatment. These promising results were obtained after administering twice weekly for 8 weeks, i.e. after a total of only 16 doses, the combinations of Q203 with either RPT or BDQ. These three drugs are long-lasting, with serum half-lives in mice after a single dose being Q203 23 h [15]; RPT 25 h [16] and BDQ 53 h [17]. Recently, regimens combining RPT with CLR or BDQ, administered twice weekly for 8 weeks, were found to be as bactericidal and as sterilizing as a daily RPT-CLR regimen [28].

We recently showed that Q203 was very active in the footpads of mice infected with M. *ulcerans* [13]. Lately, a study by Converse *et al* confirmed this finding and furthermore showed that triple combinations of Q203 administered at the higher dose of 10 mg/kg, either with RPT and clofazimine, RPT and BDQ or BDQ and clofazimine, as well as a quadruple combination of these four drugs, were sterilizing after 2 weeks of daily treatment in a BU animal model [14], leading to the conclusion that targeting the *M. ulcerans* respiratory chain with several drugs is an effective strategy for designing new shortened treatments of BU. Thus, Q203 appears clearly an important candidate for future treatment of BU, either in daily or intermittent regimens.

The fact that, in the present work, few bacilli were still found by culture in 1 out of 8 mice after 8 weeks of treatment with RIF-CLR 5/7, and that 8 out of 26 mice relapsed with low CFU counts within 20 weeks after the end of this regimen, was surprising since bactericidal and sterilizing activity of RIF-CLR was shown in our previous studies. Other authors have observed, 20 weeks after stopping treatment with RIF-CLR 5/7, 25% of relapses based on recurrent swelling or ectopic lesions [29]. The susceptibility to RIF and to CLR of the bacilli isolated from relapsing mice in our study was unchanged, ruling out the selection of resistant mutants during treatment. Relapses could be explained by unusually high bacterial loads reached in the present work when compared to those in our previous studies, *i.e.* 3–4 times higher at the start of treatment and 4–10 times higher after 2–4 weeks in untreated mice. Nevertheless, this fact strengthens further the good results obtained with the Q203-containing combination regimens.

New drugs are rare for the treatment of BU. The last active new marketed drug was BDQ, initially assessed with success for tuberculosis and found later to be very active against *M*. *ulcerans* in a BU animal model [28]. Therefore, the excellent *in vivo* activity of Q203 against *M*. *ulcerans* constitutes a step forward. In a recent study, the basis of this promising result was found to be reductive evolution in most strains of this species that led to hyper susceptibility to Q203 by elimination due to the absence of alternate terminal electron acceptors, thereby making the target, respiratory cytochrome *bc*1:*aa*3, crucial for survival [13].

Whereas, triple or quadruple combinations of Q203 at 10mg/kg, administered with RPT, clofazimine and BDQ were found to sterilize infected mice after 2 weeks of daily treatment, we demonstrated in the present study that double combinations of Q203 at 5 mg/kg with either RPT or BDQ, were also sterilizing after 16 doses of a twice weekly treatment, due to the long half-life and good bioavailability of these three drugs. The latter intermittent scheme of oral treatment with two drugs would greatly simplify the treatment of BU on ambulatory care in

the field, when patients living in remote areas visit healthcare centers twice or three times per week for dressing changes, a rhythm that could allow receiving supervised intermittent antibiotic administration.

Author Contributions

Conceptualization: Aurélie Chauffour, Vincent Jarlier.

Formal analysis: Jérôme Robert, Vincent Jarlier.

Funding acquisition: Jérôme Robert, Nicolas Veziris, Alexandra Aubry, Vincent Jarlier.

Methodology: Aurélie Chauffour, Jérôme Robert, Vincent Jarlier.

Project administration: Alexandra Aubry.

Resources: Aurélie Chauffour.

Software: Aurélie Chauffour.

Supervision: Vincent Jarlier.

Validation: Jérôme Robert, Kevin Pethe, Vincent Jarlier.

Writing - original draft: Aurélie Chauffour, Vincent Jarlier.

Writing - review & editing: Jérôme Robert, Nicolas Veziris, Alexandra Aubry, Kevin Pethe.

References

- WHO. Treatment of Mycobacterium ulcerans disease (Buruli ulcer) Guidance for health workers. 2012. Available: https://apps.who.int/iris/bitstream/handle/10665/77771/9789241503402_eng.pdf; jsessionid=AECF9887A2910FB44C46C2212817A90F?sequence=1
- Ji B, Chauffour A, Robert J, Jarlier V. Bactericidal and sterilizing activities of several orally administered combined regimens against Mycobacterium ulcerans in mice. Antimicrob Agents Chemother. 2008; 52: 1912–1916. https://doi.org/10.1128/AAC.00193-08 PMID: 18391038
- Chauty A, Ardant M-F, Marsollier L, Pluschke G, Landier J, Adeye A, et al. Oral treatment for Mycobacterium ulcerans infection: results from a pilot study in Benin. Clin Infect Dis. 2011; 52: 94–96. https://doi.org/10.1093/cid/ciq072 PMID: 21148526
- Phillips RO, Robert J, Abass KM, Thompson W, Sarfo FS, Wilson T, et al. Rifampicin and clarithromycin (extended release) versus rifampicin and streptomycin for limited Buruli ulcer lesions: a randomised, open-label, non-inferiority phase 3 trial. Lancet Lond Engl. 2020; 395: 1259–1267. https://doi.org/10. 1016/S0140-6736(20)30047-7 PMID: 32171422
- Lim LE, Vilcheze C, Ng C, Jacobs WR, Ramon-Garcia S, Thompson CJ. Anthelmintic Avermectins Kill Mycobacterium tuberculosis, Including Multidrug-Resistant Clinical Strains. Antimicrob Agents Chemother. 2013; 57: 1040–1046. https://doi.org/10.1128/AAC.01696-12 PMID: 23165468
- Omansen TF, Porter JL, Johnson PDR, van der Werf TS, Stienstra Y, Stinear TP. In-vitro Activity of Avermectins against Mycobacterium ulcerans. Johnson C, editor. PLoS Negl Trop Dis. 2015; 9: e0003549. https://doi.org/10.1371/journal.pntd.0003549 PMID: 25742173
- Scherr N, Pluschke G, Thompson CJ, Ramón-García S. Selamectin Is the Avermectin with the Best Potential for Buruli Ulcer Treatment. Johnson C, editor. PLoS Negl Trop Dis. 2015; 9: e0003996. <u>https://doi.org/10.1371/journal.pntd.0003996</u> PMID: 26270480
- Molina-Torres CA, Barba-Marines A, Valles-Guerra O, Ocampo-Candiani J, Cavazos-Rocha N, Pucci MJ, et al. Intracellular activity of tedizolid phosphate and ACH-702 versus Mycobacterium tuberculosis infected macrophages. Ann Clin Microbiol Antimicrob. 2014; 13: 13. https://doi.org/10.1186/1476-0711-13-13 PMID: 24708819
- Ji B, Lefrancois S, Robert J, Chauffour A, Truffot C, Jarlier V. In Vitro and In Vivo Activities of Rifampin, Streptomycin, Amikacin, Moxifloxacin, R207910, Linezolid, and PA-824 against Mycobacterium ulcerans. Antimicrob Agents Chemother. 2006; 50: 1921–1926. <u>https://doi.org/10.1128/AAC.00052-06</u> PMID: 16723546
- Burdette SD, Trotman R. Tedizolid: The First Once-Daily Oxazolidinone Class Antibiotic. Saravolatz LD, editor. Clin Infect Dis. 2015; 61: 1315–1321. https://doi.org/10.1093/cid/civ501 PMID: 26105167

- Makarov V, Lechartier B, Zhang M, Neres J, van der Sar AM, Raadsen SA, et al. Towards a new combination therapy for tuberculosis with next generation benzothiazinones. EMBO Mol Med. 2014; 6: 372– 383. https://doi.org/10.1002/emmm.201303575 PMID: 24500695
- Makarov V, Manina G, Mikusova K, Mollmann U, Ryabova O, Saint-Joanis B, et al. Benzothiazinones Kill Mycobacterium tuberculosis by Blocking Arabinan Synthesis. Science. 2009; 324: 801–804. https:// doi.org/10.1126/science.1171583 PMID: 19299584
- Scherr N, Bieri R, Thomas SS, Chauffour A, Kalia NP, Schneide P, et al. Targeting the Mycobacterium ulcerans cytochrome bc1:aa3 for the treatment of Buruli ulcer. Nat Commun. 2018; 9: 5370. https://doi. org/10.1038/s41467-018-07804-8 PMID: 30560872
- Converse PJ, Almeida DV, Tyagi S, Xu J, Nuermberger EL. Shortening Buruli Ulcer Treatment with Combination Therapy Targeting the Respiratory Chain and Exploiting *Mycobacterium ulcerans* Gene Decay. Antimicrob Agents Chemother. 2019; 63: e00426–19, /aac/63/7/AAC.00426-19.atom. https:// doi.org/10.1128/AAC.00426-19 PMID: 31036687
- Pethe K, Bifani P, Jang J, Kang S, Park S, Ahn S, et al. Discovery of Q203, a potent clinical candidate for the treatment of tuberculosis. Nat Med. 2013; 19: 1157–1160. <u>https://doi.org/10.1038/nm.3262</u> PMID: 23913123
- Rosenthal IM, Williams K, Tyagi S, Peloquin CA, Vernon AA, Bishai WR, et al. Potent Twice-Weekly Rifapentine-containing Regimens in Murine Tuberculosis. Am J Respir Crit Care Med. 2006; 174: 94– 101. https://doi.org/10.1164/rccm.200602-280OC PMID: 16574936
- Rouan M-C, Lounis N, Gevers T, Dillen L, Gilissen R, Raoof A, et al. Pharmacokinetics and Pharmacodynamics of TMC207 and Its N-Desmethyl Metabolite in a Murine Model of Tuberculosis. Antimicrob Agents Chemother. 2012; 56: 1444–1451. https://doi.org/10.1128/AAC.00720-11 PMID: 22155815
- Shepard CC. The experimental disease that follows the injection of human leprosy bacilli into foot-pads of mice. J Exp Med. 1960; 112: 445–454. https://doi.org/10.1084/jem.112.3.445 PMID: 19867175
- Dega H, Bentoucha A, Robert J, Jarlier V, Grosset J. Bactericidal Activity of Rifampin-Amikacin against Mycobacterium ulcerans in Mice. Antimicrob Agents Chemother. 2002; 46: 3193–3196. https://doi.org/ 10.1128/aac.46.10.3193-3196.2002 PMID: 12234844
- Inderlied CB, Nash KA. Antimicrobial agents: in vitro susceptibility testing, spectra of activity, mechanisms of action and resistance, and assays for activity in biologic fluids. 4th ed. Antibiotics in laboratory medicine. 4th ed. The Williams & Wilkins Co., Baltimore. Md: M. D. V. Lorian; 1996. pp. 127–175.
- Wadagni AC, Steinhorst J, Barogui YT, Catraye PM, Gnimavo R, Abass KM, et al. Buruli ulcer treatment: Rate of surgical intervention differs highly between treatment centers in West Africa. PLoS Negl Trop Dis. 2019; 13: e0007866. https://doi.org/10.1371/journal.pntd.0007866 PMID: 31658295
- Wadagni AC, Barogui YT, Johnson RC, Sopoh GE, Affolabi D, van der Werf TS, et al. Delayed versus standard assessment for excision surgery in patients with Buruli ulcer in Benin: a randomised controlled trial. Lancet Infect Dis. 2018; 18: 650–656. <u>https://doi.org/10.1016/S1473-3099(18)30160-9</u> PMID: 29605498
- Geyer J, Gavrilova O, Petzinger E. Brain penetration of ivermectin and selamectin in *mdr1a,b* P-glycoprotein- and *bcrp*—deficient knockout mice. J Vet Pharmacol Ther. 2009; 32: 87–96. https://doi.org/10. 1111/j.1365-2885.2008.01007.x PMID: 19161460
- Lepak AJ, Marchillo K, Pichereau S, Craig WA, Andes DR. Comparative pharmacodynamics of the new oxazolidinone tedizolid phosphate and linezolid in a neutropenic murine Staphylococcus aureus pneumonia model. Antimicrob Agents Chemother. 2012; 56: 5916–5922. https://doi.org/10.1128/AAC. 01303-12 PMID: 22964254
- Kanafani ZA, Corey GR. Tedizolid (TR-701): a new oxazolidinone with enhanced potency. Expert Opin Investig Drugs. 2012; 21: 515–522. https://doi.org/10.1517/13543784.2012.660250 PMID: 22324973
- Flanagan S, Fang E, Muñoz KA, Minassian SL, Prokocimer PG. Single- and Multiple-Dose Pharmacokinetics and Absolute Bioavailability of Tedizolid. Pharmacother J Hum Pharmacol Drug Ther. 2014; 34: 891–900. https://doi.org/10.1002/phar.1458 PMID: 24989138
- Incandela ML, Perrin E, Fondi M, de Jesus Lopes Ribeiro AL, Mori G, Moiana A, et al. DprE1, a new taxonomic marker in mycobacteria. FEMS Microbiol Lett. 2013; 348: 66–73. https://doi.org/10.1111/1574-6968.12246 PMID: 24024613
- Chauffour A, Robert J, Veziris N, Aubry A, Jarlier V. Sterilizing Activity of Fully Oral Intermittent Regimens against Mycobacterium Ulcerans Infection in Mice. Small PLC, editor. PLoS Negl Trop Dis. 2016; 10: e0005066. https://doi.org/10.1371/journal.pntd.0005066 PMID: 27755552
- Converse PJ, Tyagi S, Xing Y, Li S-Y, Kishi Y, Adamson J, et al. Efficacy of Rifampin Plus Clofazimine in a Murine Model of Mycobacterium ulcerans Disease. Phillips RO, editor. PLoS Negl Trop Dis. 2015; 9: e0003823. https://doi.org/10.1371/journal.pntd.0003823 PMID: 26042792